Author's response to reviews

Title: Factors Associated with Adoption of Beneficial Newborn Care Practices in Rural Eastern Uganda: a Cross-sectional Study

Authors:

Michael Odoi Owor Dr (owormichael@gmail.com)
Joseph KB Matovu (jmatovu@musph.ac.ug)
Daniel Murokora Dr (murokora@gmail.com)
Rhoda K Wanyenze Dr (rwanyenze@musph.ac.ug)
Peter Waiswa Dr (pwaiswa@musph.ac.ug)

Version: 9 Date: 18 March 2016

Author's response to reviews: see over
March 18, 2016

The Editor,
BMC Pregnancy and Childbirth

Dear Editor,

**Re: Response to Reviewers Comments for Manuscript#: 1261033346145405**

We remain grateful to the reviewers’ and the editor who re-read and provided the latest comments and guidance to our manuscript mentioned above.

We remain thankful to you yet again for guiding us intuitively to improve the manuscript further in the light of the comments made.

In this version of the manuscript, we have addressed the comments and we thank you for the effort to review our manuscript again.

Below is a point-by-point response to the comment(s) raised.

We look forward to receiving your further guidance on this subject.

Regards,

Owor Michael Odoi
Corresponding Author
## POINT-BY-POINT RESPONSE TO REVIEWER’S COMMENTS

<table>
<thead>
<tr>
<th>Comments</th>
<th>Response</th>
</tr>
</thead>
<tbody>
<tr>
<td>Reviewer 1</td>
<td>Role of DM</td>
</tr>
</tbody>
</table>

Dear editor,

DM (Dan Murokora), alongside JKBM, RW, provided guidance in the review of the manuscript.

Additionally, Dr. Fredrick Makumbi is acknowledged for providing guidance in the statistical analysis of data as opposed to writing the manuscript as stated in the earlier versions of the manuscript. This change has been made and highlighted in yellow.